Improving the diagnosis and management of Lewy body dementia: the DIAMOND-Lewy research programme including pilot cluster RCT

John T O'Brien,^{1,2*} John-Paul Taylor,² Alan Thomas,² Claire Bamford,³ Luke Vale,³ Sarah Hill,³ Louise Allan,⁴ Tracy Finch,⁵ Richard McNally,³ Louise Hayes,³ Ajenthan Surendranathan,¹ Joseph Kane,² Alexandros E Chrysos,³ Allison Bentley,¹ Sally Barker,² James Mason,⁶ David Burn³ and Ian McKeith²

¹Department of Psychiatry, University of Cambridge, Cambridge, UK

²Translational and Clinical Research Institute, Newcastle University, Newcastle upon Tyne, UK

³Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK ⁴Institute of Health Research, University of Exeter, Exeter, UK

⁵Department of Nursing, Midwifery and Health, Faculty of Health and Life Sciences, Northumbria University, Newcastle upon Tyne, UK

⁶Warwick Medical School, University of Warwick, Coventry, UK

*Corresponding author john.obrien@medschl.cam.ac.uk

Declared competing interests of authors: John T O'Brien reports grants from the National Institute for Health Research (NIHR) during the conduct of the study, personal fees from TauRx Therapeutics Ltd (Aberdeen, UK), grants and personal fees from Avid Radiopharmaceuticals/Eli Lilly and Company (Indianapolis, IN, USA) and personal fees from Eisai Co. Ltd (Tokyo, Japan), GE Healthcare (Chicago, IL, USA), and AXON Neuroscience (Bratislava, Slovakia) outside the submitted work. John-Paul Taylor reports grants from NIHR during the conduct of the study, non-financial support from Axovant Gene Therapies Ltd (Hamilton, Bermuda) and personal fees from GE Healthcare outside the submitted work. Alan Thomas reports grants from NIHR during the conduct of the study, and grants from GE Healthcare outside the submitted work. Luke Vale reports grants from NIHR during the conduct of the study, and membership of the Health Technology Assessment (HTA) Clinical Trial Committee (2014–18). Richard McNally reports grants from NIHR during the conduct of the study, and membership of the Health Services and Delivery Research (HSDR) Commissioned Panel (2015–19). James Mason reports grants from NIHR during the conduct of the study, and membership of the NIHR HSDR Funding Committee (2017–20), the NIHR HTA End of Life Care and Add-on Studies board (2015-2016) and the NIHR HTA Efficient Study Designs - 2 board (2015-16). David Burn reports grants from NIHR during the conduct of the study, and membership of the Efficacy and Mechanism Evaluation Prioritisation Group (2012–16). Ian McKeith reports grants from NIHR during the conduct of the study, personal fees from Eisai Co. Ltd, GE Healthcare, Axovant Gene Therapies Ltd, Sumitomo Dainippon Pharma (Osaka, Japan) and Sanofi SA (Paris, France), and grants and personal fees from La Fondation pour la Recherche sur Alzheimer (Paris, France) outside the submitted work.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published July 2021 DOI: 10.3310/pgfar09070

Plain English summary

The DIAMOND-Lewy research prgramme Programme Grants for Applied Research 2021; Vol. 9: No. 7 DOI: 10.3310/pgfar09070

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Lewy body dementia includes dementia with Lewy bodies and Parkinson's disease dementia. It is the second most common cause of neurodegenerative dementia in older people, but is underdiagnosed. There is no consistent approach to management.

We aimed to improve the diagnosis and management of Lewy body dementia by:

- understanding current practice
- identifying barriers to diagnosis and management
- developing evidence-based assessment toolkits
- producing and piloting a management toolkit.

We focused on hospital-based specialist services. Participants included people aged \geq 50 years diagnosed with Lewy body dementia, carers and health-care professionals. Current diagnostic rates for dementia with Lewy bodies and Parkinson's disease dementia were lower than expected (4.6% and 9.7%, respectively). There was evidence of delays in diagnosis for both dementia with Lewy bodies and Parkinson's disease dementia costs of care for patients with dementia with Lewy bodies and Parkinson's disease dementia were greater than for patients with dementia with Lewy bodies and Parkinson's disease dementia were greater than for patients who had other dementias. We found many barriers to the diagnosis and management of Lewy body dementia and these indicated the need to upskill staff, challenge negative attitudes towards diagnosis and improve management pathways.

Using published evidence and clinical opinion, we produced assessment and management toolkits for dementia with Lewy bodies and Parkinson's disease dementia. We used these in a pilot clinical study and showed that we were able to recruit people with dementia with Lewy bodies and Parkinson's disease dementia to the study. The toolkits were well received, but their use by clinical staff was variable. Following introduction of the assessment toolkits, there was an increase in diagnostic rates for dementia with Lewy bodies, but not Parkinson's disease dementia. Our findings indicate that a further larger study of the management toolkit is both feasible and merited.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number DTC-RP-PG-0311-12001. The contractual start date was in January 2014. The final report began editorial review in May 2019 and was accepted for publication in March 2021. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2021 O'Brien *et al.* This work was produced by O'Brien *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk